

G OPEN ACCESS

Citation: Tang B, Ma K, Liu Y, Wang X, Tang S, Xiao Y, et al. (2024) Managing spatio-temporal heterogeneity of susceptibles by embedding it into an homogeneous model: A mechanistic and deep learning study. PLoS Comput Biol 20(9): e1012497. https://doi.org/10.1371/journal. pcbi.1012497

Editor: Kimberly M. Pepin, Animal and Plant Health Inspection Service, UNITED STATES OF AMERICA

Received: April 16, 2024

Accepted: September 17, 2024

Published: September 30, 2024

Copyright: © 2024 Tang et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data and code used are available at GitHub at: https://github.com/ yanliu0321/Managing-spatial-temporalheterogeneity-of-susceptibles.git.

Funding: YX was funded by the National Key R&D Program of China (2022YFA1003704) and Major International (Regional) Joint Research Project of the National Natural Science Foundation of China (12220101001). BT was funded by the National Natural Science Foundation of China (12371502, RESEARCH ARTICLE

Managing spatio-temporal heterogeneity of susceptibles by embedding it into an homogeneous model: A mechanistic and deep learning study

Biao Tang^{1,2}, Kexin Ma¹, Yan Liu¹, Xia Wang³, Sanyi Tang³, Yanni Xiao^{1,2}*, Robert A. Cheke^{4,5}*

1 School of Mathematics and Statistics, Xi'an Jiaotong University, Xi'an, People's Republic of China, 2 The Interdisciplinary Research Center for Mathematics and Life Sciences, Xi'an Jiaotong University, Xi'an, People's Republic of China, 3 School of Mathematics and Statistics, Shaanxi Normal University, Xi'an, People's Republic of China, 4 Natural Resources Institute, University of Greenwich at Medway, Central Avenue, Chatham Maritime, Kent, United Kingdom, 5 Department of Infectious Disease Epidemiology, Imperial College London, School of Public Health, White City Campus, London, United Kingdom

* yxiao@mail.xjtu.edu.cn (YX); r.a.cheke@greenwich.ac.uk (RAC)

Abstract

Accurate prediction of epidemics is pivotal for making well-informed decisions for the control of infectious diseases, but addressing heterogeneity in the system poses a challenge. In this study, we propose a novel modelling framework integrating the spatio-temporal heterogeneity of susceptible individuals into homogeneous models, by introducing a continuous recruitment process for the susceptibles. A neural network approximates the recruitment rate to develop a Universal Differential Equations (UDE) model. Simultaneously, we pre-set a specific form for the recruitment rate and develop a mechanistic model. Data from a COVID Omicron variant outbreak in Shanghai are used to train the UDE model using deep learning methods and to calibrate the mechanistic model using MCMC methods. Subsequently, we project the attack rate and peak of new infections for the first Omicron wave in China after the adjustment of the dynamic zero-COVID policy. Our projections indicate an attack rate and a peak of new infections of 80.06% and 3.17% of the population, respectively, compared with the homogeneous model's projections of 99.97% and 32.78%, thus providing an 18.6% improvement in the prediction accuracy based on the actual data. Our simulations demonstrate that heterogeneity in the susceptibles decreases herd immunity for ~37.36% of the population and prolongs the outbreak period from ~30 days to ~70 days, also aligning with the real case. We consider that this study lays the groundwork for the development of a new class of models and new insights for modelling heterogeneity.

Author summary

This study developed a novel modelling framework that explicitly integrates spatial heterogeneity into traditionally homogeneous compartmental models for precise and 12101488), and was supported by the Young Talent Support Plan of Xi'an Jiaotong University. ST was funded by the National Natural Science Foundation of China (12031010). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

efficient prediction of epidemics of infectious diseases. We employed the deep learning method to learn the recruitment rate of susceptible individuals from data, given its inherent complexity and limited information. Moreover, we predefined the mechanistic form of the recruitment rate to facilitate interpretation and scenario analysis, where we fitted the recruitment rate of the mechanistic model to those obtained from the deep learningbased approach. Compared to the homogeneous model, our mechanistic model predicts significantly lower peak and final sizes of the epidemic, as well as reduced levels of herd immunity. Using real data as a benchmark, our model's predictive accuracy has markedly improved. The framework of our mechanistic model employs a straightforward approach to integrate heterogeneity into compartment models, thereby facilitating the extension of the modelling framework to encompass various aspects of infectious disease transmission with spatio-temporal heterogeneity among the susceptible population.

Introduction

Over the past few decades, emerging infectious diseases, such as SARS, influenza A H1N1, and SARS-CoV-2, have posed significant threats to global public health [1-3], and the control of such infectious diseases has garnered considerable attention from researchers [4,5]. Mathematical models play a crucial role as tools for understanding the transmission mechanisms of emerging infectious diseases, predicting epidemic trends, and evaluating transmission risks. These models can aid in optimizing control strategies to enhance global public health and reduce economic costs [6-8]. In particular, the precise prediction of epidemic trends serves as a foundational basis for decision-making in the implementation of control interventions. However, there are numerous challenges associated with achieving accurate predictions, with heterogeneity being one of the significant obstacles to achieving precise predictions.

A susceptible population is usually defined as comprising those individuals who can be infected by a virus or other pathogen. The transmission of pathogens occurs through several mechanisms: one is direct transmission involving physical contact between infectious individuals and the susceptible population, as seen with many respiratory viruses. In another mechanism, unique to vector-borne diseases transmission occurs when an infected vector, such as a mosquito, bites a susceptible human. Sexually transmitted diseases are spread through sexual contact, which can include any form of sexual activity involving the exchange of body fluids. However, regardless of the transmission mechanism, the simplest SIR model structure assumes that all susceptibles are equally likely to be infected by infectious individuals, even though many of them may not have the opportunity to come into contact with an infectious individual simultaneously due to a heterogeneous contact structure or spatial processes. Therefore, when considering the spatio-temporal heterogeneity of susceptibles, the actual susceptible population in homogeneous models should consist of individuals actively involved in the transmission process—those with an approximately equal probability of encountering infectious individuals. Moreover, additional susceptible individuals may have increased contact with infectious individuals as the epidemic spreads spatially. This phenomenon is intuitively observed in nearly all outbreaks of SARS-CoV-2 (henceforth COVID-19) and other infectious diseases. For instance, Fig 1 illustrates the spatio-temporal shift of confirmed cases during a local outbreak of the Omicron variant of COVID-19 in Shanghai, China, from 1 to 17 March 2022. In this scenario, the Chongming district of Shanghai reported its first case on 17 March 2022-17 days after the diagnosis of the initial confirmed case in the Putuo district on 1 March 2022. This observation underscores the dynamic nature of the susceptible population and its



Fig 1. Heat-map of the cumulative numbers of infected cases showing the spatial spread of the Omicron variant of COVID-19 during the initial phase of the outbreak in Shanghai city during 1–17 March 2022. The source of the basemap shapefile was from the open access platform: National Platform for Common Geospatial Information Services (https://www.tianditu.gov.cn/; note that this link works well in P.R. China, but may be blocked in some other countries).

involvement in the transmission process as infections spread over space and time. The significance of spatial heterogeneity has long been acknowledged in epidemic modelling, highlighting the limitations of homogeneous assumptions [9,10]. Consequently, numerous studies have explored various methods to incorporate spatial heterogeneity, yielding a wealth of intriguing findings [11–14].

Patch, network and spatial reaction-diffusion models can serve as frameworks for modelling the intricacies of spatial heterogeneity. Numerous existing studies have demonstrated that these modelling frameworks qualitatively enhance our understanding of the underlying transmission mechanisms of infectious diseases [11,12,15,16]. Nevertheless, there are challenges in establishing connections between these models and real-world data, conducting quantitative analyses, and making predictions due to their inherent structural complexity. Specifically, there is a scarcity of epidemic data with high-resolution spatial information, or the available data exhibit a granularity that is too coarse to effectively capture spatial nuances. Additionally, fitting epidemic data to network models or spatial diffusion equations proves to be highly challenging owing to the intricate nature of the model structures. Furthermore, the complexity of these model structures hinders the comprehensive analysis or prediction of fundamental epidemic indices, including peak values, final sizes and herd immunity.

For this reason, homogeneous compartment models are frequently used in scenario analysis to predict the epidemic trends of emerging infectious diseases due to the huge advantages of this modelling framework, given the aspects of parameterization, interpretability, ease of calibration, computational efficiency and strong predictive capabilities. Although spatial heterogeneity has been recognized in epidemic modelling for decades, most research has focused on modifying the transmission terms within homogeneous compartmental models to incorporate spatial heterogeneity. Consequently, various types of incidence rates have been proposed, revealing a range of complex dynamic behaviours [10,17,18]. However, the transmission terms describe a process in which there is no explicit spatial behaviour (no 'movement parameters'), although spatial distribution and movement is implicitly incorporated. Subsequently, in various scenarios, the prediction accuracy of the existing homogeneous models may be compromised by oversimplification, as they tend to disregard important heterogeneity factors. Several recent studies [19,20] have developed explicit methods that incorporate spatial heterogeneity to quantify the spatio-temporal transmission of wildlife diseases. However, these methods require high-resolution spatial data. Therefore, exploring novel approaches that explicitly integrate spatial heterogeneity within homogeneous compartmental modelling frameworks to add new insights for modelling heterogeneity, particularly for precise (by involving heterogeneity) and efficient (by leveraging the advantages of homogeneous model framework) prediction of epidemics, is the scope of this study.

The main purpose of this study is to develop a generalized modelling framework that embeds spatio-temporal heterogeneity of the susceptible population into the homogeneous models. The ultimate aim is to leverage the homogeneous compartment model for conducting efficient prediction of epidemic trends, and simultaneously enhance the prediction accuracy of the epidemics. In the next section, we firstly develop a generalized framework, and design a machine-learning approach and pre-set a specific form to shape the recruitment process from the epidemic data of COVID-19. Subsequently, we apply our proposed model to project the epidemic trends of various COVID-19 variants as case studies, conducting tests and evaluations to compare prediction accuracy with traditional homogeneous modelling frameworks. Finally, we discuss the biological implications and the prospective applications of our novel modelling framework.

Methods

Model formulation

The classical *SIR* compartment model proposed by Kermack and McKendrick in 1927 [21,22] is:

$$\begin{cases} S' = -\beta_0 c I \frac{S}{N}, \\ I' = \beta_0 c I \frac{S}{N} - \gamma I, \\ R' = \gamma I, \end{cases}$$
(1)

where *S*, *I*, *R* denote the susceptible, infectious, and recovered populations, respectively, N = S + I + R denotes the total population in the considered region, which is constant as dN/dt = 0. Here, we consider the variables as the absolute population while the units are listed in Table 1. The parameter γ is the recovery rate. The term $\beta_0 c \frac{S}{N} I$ is the standard incidence rate for describing the transmission of the disease. That is, each infectious individual can generate a constant number of contacts *c* per unit time with a transmission probability β_0 per contact, correspondingly, $\beta = \beta_0 c$ is usually called the transmission rate. In addition, *S*/*N* is the probability at which each infectious individual contacts with susceptible individuals, because only the contacts between infectious and susceptible individuals can transmit the pathogen.

	Definition	Values: mean (95% CI)		Units	Source
		UDE Model	Mechanistic model		
Paramet	ers				
β	$\beta = \beta_0 c$ (transmission rate)	0.4743	0.4945 (0.4943, 0.4946)	Day ⁻¹	Estimated
β_0	Transmission probability per contact				
с	Constant contact rate per day			Day ⁻¹	
γ	Recovery rate	1/5		Day ⁻¹	Ref [35]
k	$k = \eta p$		0.2812 (0.2810, 0.2814)		Estimated
р	Probability that the contact happens outside the epicentre				
η	Proportionality factor				
R ₀	Initial (basic) reproduction number	2.37	2.47		Estimated
Variable	s				
N(0)	Total population	$2.475^{*}10^{7}$		People	Database
$S_r(0)$	Initial reserve susceptible population	$N(0) - S_e(0)$		People	Calculated
<i>S</i> _e (0)	Initial effective susceptible population	400002	409999 (409997, 410001)	People	Estimated
I(0)	Initial infections	1.778	0.7041 (0.7005, 0.7078)	People	Estimated
R(0)	Initial recovered populations	0	0	People	Assumed

Table 1.	Definitions and	values of	parameters and	variables ir	n model (3) and model	(4)
1.0010 11			P			/	· · /

The fundamental premise underlying homogeneous models is that the population is homogeneously mixing, so the proportion of infectious individuals around any susceptible individual is equal regardless of the location of the individual in the area. Consequently, when considering a single new outbreak of an emerging infectious disease, the entire population is considered as the initial condition for the susceptible class, i.e. (S(0) = N). Over the past few decades, the SIR model has undergone extensive extensions to various frameworks, such as SEIR and SEIHR models. However, most of these extensions have retained the assumptions related to managing the susceptible population [23–27]. As discussed in the introduction, such assumptions are deemed unreasonable due to their overlooking of spatial distances between individuals.

To establish a more realistic modelling framework, we posit that transmission only occurs among the population within a small neighbourhood of all the infectious individuals, who are also assumed to be uniformly mixed within the neighbourhood. Therefore, the transmission among the population within the neighbourhood can be appropriately modelled within the homogeneous compartment model framework, specifically concerning the transmission of the pathogen through physical contacts. In this context, an area under consideration can be spatially divided into two distinct regions: the neighbourhood of all the infections, referred to as the epicentre, and the remaining region outside the epicentre (RoEC), as illustrated in Fig 2. As depicted in Fig 1, the evolution of diseases reveals the spatial diffusion of infected cases since the initiation of the outbreak. Therefore, the transmission should be initially considered within a small region that covers all the infectious individuals, i.e. the epicentre, and this will be expanded as the spatial diffusion of infectious individuals progresses. That is, attributed to the spatial diffusion of infectious individuals, the infections in the epicentre can randomly contact with the susceptible population in RoEC, the region of the epicentre should then be expanded and consequently more susceptibles will be involved in the transmission process, as



Fig 2. Schematic diagram illustrating the recruitment process of susceptible individuals. Within the epicentre, the classic SIR compartment model remains applicable for describing the transmission of the pathogen. The crucial aspect lies in incorporating the recruitment process of the susceptible population into the classic SIR compartment model, recognizing the spatio-temporal heterogeneity of the susceptibles. We term the susceptible individuals in the epicentre as the effective susceptible population, denoted by S_{e^*} and those in RoEC are referred to as the reserved susceptible population, denoted by S_r . Considering the continuous spatial spread of infections, the epicentre is expanded to encompass more susceptibles. As a result, the susceptible population in class S_r will transition into the newly added epicentre and become S_e . Letting f(t) represent the transition rate from $S_r(t)$ to $S_e(t)$, we establish the following generalized modelling framework in terms of the transition process of susceptibles.

shown in Fig 2. This is in line with the modelling idea of reaction-diffusion epidemic models with free boundary [28, 29].

$$\begin{cases} S'_r = -f(t), \\ S'_e = f(t) - \frac{\beta S_e I}{N_e}, \\ I' = \frac{\beta S_e I}{N_e} - \gamma I, \\ R' = \gamma I. \end{cases}$$
(2)

where $N = S_r + S_e + I + R$ denotes the total population in the considered region, which is a constant as dN/dt = 0. The population involved in the transmission process at time *t*, i.e. $N_e(t) = S_e(t) + I(t) + R(t)$, is assumed to be uniformly mixed, hence we use the standard incidence rate $\frac{\beta S_e I}{N_e}$ to describe the transmission process of the pathogen between infectious individuals and susceptible individuals.

Estimation of the transition rate of f(t)

Next, we focus on determining and estimating f(t). Susceptibles are involved in the transmission process because the infectious individuals make contact with the susceptible population outside the epicentres. Therefore, it is reasonable to assume that a larger population of infectious individuals should have a higher probability of contacting individuals in RoECs,

consequently the transition rate at time *t* (i.e. f(t)) should be proportional to the current infections (i.e. I(t)), that is $f(t) \propto I(t)$.

Several advanced methods, such as Universal Differential Equations (UDEs) [30] and Physics-Informed Neural Networks (PINNs) [31], have been developed to integrate mechanistic models with neural networks. These well-established and extensively validated components are directly incorporated into the mechanistic model, while the neural network is employed to approximate complex and less-well-understood functions. f(t) should have a complex form but there is very limited information and existing experience on the form of the function. Therefore, the approach coupling the mechanistic model with neural networks is an attractive way to shape the transition rate from data directly. It should be noticed that PINN introduce physical laws into the training process as part of the loss function, while it uses neural networks to approximate and solve the variables (such as S(t), I(t), R(t) in this study). In contrast, UDEs, which originate from neural ordinary differential equations, maintain the compartmental frameworks but introduce a neural network solely to approximate unknown functions, thus staying much closer to a traditional compartmental modelling framework. For this reason, we initially employ the UDE framework and the technology of deep learning to shape the form of f(t) from epidemic data.

The approach of UDEs helps to avoid unreasonable assumptions by allowing the neural networks to learn functions directly from the data, however, it has a poor performance when attempting generalization to conduct scenario analysis under different situations due to the lack of a mechanistic form. Therefore, it is also essential to find a special form of the function, which should be also in line with the shape learned from UDEs. Both symbolic regression and cutting-edge neural networks (Kolmogorov-Arnold Networks (KANs)) [32] provide feasible methods to learn a mechanistic form, however, both of these methods have a poor robustness when reconstructing the functions, and such reconstructed functions are usually complex, hence not intuitive. Alternatively, we try to pre-set the specific form of f(t) and estimate the parameters in f(t) by fitting the model to the epidemic data and the output data of f(t) by the UDEs, simultaneously.

Universal differential equation-based model

Here, we present a data-driven approach designed to utilize epidemic data for shaping the time-varying recruitment rate of susceptibles. As mentioned above, given the limited information on the transition rate, the UDE modelling framework can be a very good choice for learning the recruitment pattern of susceptibles from epidemic data. That is, utilizing the universal approximation property of deep neural networks, we use a neural network to present f(t) in the absence of the specific form of it, and then train the neural network to shape the specific form of f(t) for particular outbreaks of infectious diseases. Incorporating the assumption of $f(t) \propto I(t)$, we therefore have $f(t) = r^{NN}(t) * I(t)$ with $r^{NN}(t)$ being the neural network to approximate the time-varying recruitment rate of the susceptible population from the RoECs to the epicentres. Consequently, we propose the following universal differential equations model based on the generalized modelling framework (model (2)) [30,33]:

$$S'_{r} = -r^{NN}(t)*I,$$

$$S'_{e} = r^{NN}(t)*I - \frac{\beta S_{e}I}{N_{e}},$$

$$I' = \frac{\beta S_{e}I}{N_{e}} - \gamma I,$$

$$R' = \gamma I.$$
(3)

By transferring the training process of the UDE model to solve an optimal control problem, we can extend the idea of back-propagation of neural networks to train the neural network included in the UDE model, in differential programming with the fully fledged technique of adjoint sensitivity analysis [34].

Mechanistic model

We also try to find a particular form of the time-varying recruitment rate for understanding the transition mechanism of the susceptible population. As mentioned above, each infectious individual can generate a constant contact rate *c* per unit time. We then set *p* as the probability of the contacts happening outside their epicentre, hence, there are in total pcI(t) contacts outside the epicentre at time *t*. Similar to the standard incidence rate, contacts should also happen between infectious individuals I(t) and the reserved susceptible population $S_r(t)$ for successfully introducing new infections in RoEC and enlarge the epicentre, and the corresponding probability is given by S_r/N . Then, it is reasonable to assume that the recruitment rate should be proportional to the newly acquired infections in the RoEC. Consequently, we have the following form of the recruitment rate:

$$f(t) = f(S_r(t), I(t)) = \eta * \beta_0 * pcI * \frac{S_r}{N} = k\beta I \frac{S_r}{N},$$

where $k = \eta p$ with η being the proportionality factor and $\beta = \beta_0 c$. Obviously, the above form of the recruitment rate satisfies the basic assumption that f(t) is proportional to the current infections I(t) with $\frac{\partial f}{\partial 1} > 0$. Further, there is $\frac{\partial f}{\partial s_r} > 0$, indicating that the recruitment rate is also increasing as there are more susceptibles in the RoEC. Then, the generalized model (2) has the following specific form

$$\begin{cases} S'_{r} = -k\beta I \frac{S_{r}}{N}, \\ S'_{e} = k\beta I \frac{S_{r}}{N} - \frac{\beta S_{e}I}{N_{e}}, \\ I' = \frac{\beta S_{e}I}{N_{e}} - \gamma I, \\ R' = \gamma I. \end{cases}$$

$$(4)$$

The definitions and values of all the parameters and variables for both the mechanistic model and the UDE model are given in Table 1.

Data and data processing

We collected the epidemic data on the local outbreak of the Omicron variant of COVID-19 starting at Shanghai city, China, from 1 March to 2 July 2022 from the Shanghai Municipal Health Commission [36]. There are 16 districts in Shanghai city, namely, Pudong, Huangpu, Xuhui, Changning, Jingan, Putuo, Hongkou, Yangpu, Minhang, Baoshang, Jiading, Jinshan, Songjiang, Qingpu, Fengxian and Chongming districts, as shown in Fig 1. Correspondingly, we obtained the time series of the numbers of daily reported cases in each district, and the time series of the total daily numbers of reported cases in the whole of Shanghai city, as shown in Fig 3a. We also obtained part of the epidemiological survey data during 1 to 9 March 2022 from the Shanghai Municipal Health Commission [36], which includes 93 confirmed cases with the report date and the date that they closely contacted infections. Consequently, we can obtain the time-interval from the date of infection (physical contact) to the date of the report



Fig 3. Data and processing data. (a) Time series of the numbers of daily reported cases of the Omicron variant in the 16 districts of Shanghai city from 1 March to 2 July 2022. (b) Frequency distribution histogram of the time-interval from infection to report and the fitting results of the probability distribution. (c) Processed data of the numbers of daily new infections.

for the 93 confirmed cases, and the frequency distribution histogram of the time-interval is shown in Fig 3b.

We then processed the data to generate the time series of the numbers of daily new infections for model calibration as we did not consider the incubation period in the proposed model. We therefore firstly use three commonly used probability distributions, i.e. Gamma, Weibull, and Log-normal distributions, to fit the probability distributions of the time-interval from infection to report, as shown in Fig 3b. With the standard of minimizing the AIC, the Log-normal distribution was selected as the best choice. Based on the above distribution, we can further use the deconvolution method to obtain the numbers of daily new infections by processing the time series of the numbers of daily reported cases [37–39]. In detail, the number of daily new infections is denoted by N_j^I , where $j = t_1, \ldots, t_k$ with k being the length of the time series. The N_j^I are independent of each other and follow the Poisson distribution with mean λ_j , i.e. $N_j^I \sim P(\lambda_j)$. The probability of the numbers of infected cases on day *j* being reported on day *i* is denoted by p_{ij} , which corresponds to the time interval from infection to report and was obtained by discretization of the Log-normal distribution as shown in Fig 3b. By incorporating the time series of the numbers of daily reported cases C_i , $i = s_1, \ldots, s_m$ (*m* is the length of the time series), we can obtain the point estimate value $\hat{\lambda}_j$ for N_j^I by iterating the following formula:

$$C_i^n = \sum_{j=t_1}^i \lambda_j^n p_{ij}, \lambda_j^{n+1} = \frac{\lambda_j^n}{q_j} \sum_{i=max\{j,S_1\}}^{S_m} \frac{p_{ij}C_i}{C_i^n},$$

where *n* is the number of iterations, $q_j = \sum_{i=\max\{j,s_1\}}^{s_m} p_{ij} > 0, j \in [t_1, t_k]$. The iterative procedure stops when the fitting error $\frac{1}{m} \sum_{i=s_1}^{s_m} \frac{(C_i^n - C_i)^2}{C_i^n}$ is small enough [37–39]. Through the above process, the processed data of the time series of the numbers of daily new infections are shown Fig 3c.

Model training and calibration

We used the numbers of daily new infections in the local outbreak of the Omicron variant in Shanghai city to train the UDE model (i.e. model (3)) and we also calibrated the mechanistic model (model (4)). To this end, we firstly fixed the recovery rate as $\gamma = 1/5$ informed from the literature [35]. Before 1 March 2022, Shanghai city mainly experienced the wild type and Delta variants of COVID-19 with cumulative infected cases being less than 0.2% of the total population [36], hence the initial number in the recovery class is assumed to be 0 when the Omicron variant circulated for the first time in March 2022 in Shanghai, and the total population of Shanghai city was $N = 2.475^*10^7$. As a consequence, we only need to estimate the number of initial infections (I(0)), the initial effective susceptible population ($S_e(0)$), the transmission rate (β) and the parameter k for the mechanistic model (model (4)), and the parameters in the neural network of the UDE model (model (3)).

To train the UDE model, the loss function is defined as:

$$\text{Loss}_{\theta} = \sum_{t=1}^{n} (\text{Data}_{t} - \int_{t-1}^{t} \frac{\beta S_{e}(t) I(t)}{N_{e}(t)})^{2},$$

where $\int_{t-1}^{t} \frac{\beta S_e(t)I(t)}{N_e(t)}$ is the number of daily new infections at time t solved by the UDE model (model (3)), θ is the number of unknown parameters to be estimated, including the parameters in the neural network, the transmission rate β , and the initial conditions of $S_e(0)$, I(0). Then, training of model (3) is actually defined as solving the following optimal control problem:

$$\begin{split} \min_{\theta} Loss_{\theta} \Leftrightarrow \min_{\theta} \sum_{t=1}^{n} \left(\text{Data}_{t} - \int_{t-1}^{t} \frac{\beta S_{e}(t) I(t)}{N_{e}(t)} \right)^{2}, \\ \text{s.t. } S_{e}(t), I(t), N_{e}(t) \text{ satisfy model } (3) \end{split}$$

Deep learning processes for training the UDE model were implemented in the open source Julia language 1.9.2. In particular, we defined a two-layer neural network, where the first layer has an input length of 1 (i.e. time *t*) and an output length of 32 and the second layer has an input length of 32 and an output length of 1 (i.e. the recruitment rate at time *t* (f(t))). We used Tanh as the activation function, and the fifth order explicit Runge-Kutta method to solve the neural network differential equations. In training the model, we set the batch size = 1 and optimized 500 times using the Adam optimizer with an initial learning rate = 0.02 due to the different lengths of the input temporal data. Then the model was optimized 2000 times using BFGS.

When fitting the data to model (4), the objective function F is defined as the residual sum of squares between the real data of the time series of the number of daily new infections, the

recruitment rate estimated by the UDE model to the correspondingly predicted numbers by solving model (4), hence:

$$F = \sum_{t=1}^{n} (Data_{t} - \int_{t-1}^{t} \frac{\beta S_{e}(t)I(t)}{N_{e}(t)})^{2} + \sum_{t=1}^{n} \left(f_{UDE}(t) - k\beta I(t) \frac{S_{r}(t)}{N(t)} \right)^{2},$$

where $\int_{t-1}^{t} \frac{\beta S_e(t)I(t)}{N_e(t)}$ is the number of daily new infections projected by the mechanistic model (model (4)) on day *t* while *Data_t* is the observed number of the data; $f_{UDE}(t)$ is the time series of the estimated recruitment rate by the UDE model; *n* denotes the sample size of the time series of the number of daily new infections. We used the adaptive Metropolis-Hastings (M–H) algorithm to carry out the Markov Chain Monte Carlo (MCMC) procedure for data fitting and parameter estimation of the mechanistic model [40]. The algorithm was run for 50,000 iterations with a burn-in of 300,000 iterations, and the Geweke convergence diagnostic method was employed to assess convergence of chains.

Results

The training results of the UDE model and the best fitting results of the mechanistic model are shown in Fig 4a. Correspondingly the estimated values of the parameters and initial conditions are detailed in Table 1. It follows from Fig 4a that both the mechanistic model and the UDE model can fit the epidemic data of the numbers of daily new infections very well. The fitted curves of these distinct models consistently maintain proximity throughout the observed



Fig 4. Model calibrations and solutions. (a) Training result of the UDE model (red dashed curve) and the best fitting result of the mechanistic model (blue solid curve), here the blue shaded area is the 95% confidence interval of the fitting result of the mechanistic model. (b) Estimated effective reproduction number from the two models. (c) Estimated recruitment rate of the susceptible population f(t). (d)-(g) Solutions of the UDE model and the mechanistic model by fixing the parameters and initial conditions as the estimated values listed in Table 1.

https://doi.org/10.1371/journal.pcbi.1012497.g004

period. Of particular significance, Fig 4c shows the functions of the recruitment rate over time in both models. It is evident that the two curves exhibit a high degree of concordance, indicating a robust fit. This alignment supports the validity of our chosen formula in the mechanistic model for characterizing the enigmatic recruitment process of the susceptible population, aligning closely with patterns discerned from the epidemic data by the machine learning methods. Similar insights emerge upon scrutiny of the solutions for the reserved susceptible population and the effective susceptible population, as depicted in Fig 4d and 4e, respectively, for both the mechanistic and UDE models.

Notably, the size of the estimated initial effective susceptible population of the mechanistic model is approximately 409,999 As the COVID-19 spread in Shanghai city, the total numbers in the susceptible population involved in the epidemic wave escalated to 842,212. This implies that only a fraction of the Shanghai city population, specifically 3.4%, were in the population of actual susceptible individuals during the outbreak, which is characterized by an accumulation of 625,437 infections in reality. Therefore, this aspect offers a practical method for estimating the size of the actual susceptible population for a specific infectious disease outbreak, where the actual susceptible population equals the entire susceptibles minus the final reserved susceptibles $S_{r\infty}$.

Subsequently, we conduct scenario analysis, in terms of k and the transmissibility β associated with distinct variants of COVID-19, to show in depth how the heterogeneity affects the transmission dynamics, by comparing the difference between the epidemic curves produced by the novel mechanistic model (model (4)) and the homogeneous model (i.e. model (1)) by choosing the same parameter values for the two models. By fixing the recovery rate $\gamma = 1/5$ for both model (1) and model (4), and choosing the basic reproduction number as 2.2 [41], 5 [42], and 8.2 [35,43] for the wild type, Delta and Omicron variants of COVID-19, the transmission rate β can be calculated as 0.44, 1 and 1.64, respectively, for the two models. The initial conditions were fixed as the same as those of the estimation results outlined in Table 1. Consequently, we present the solutions of model (4) for different variants with varying basic reproduction numbers in Fig 5. Additionally, Table 2 compiles the corresponding peak size, attack rate, final size, and herd immunity for each scenario.

Based on the findings from Table 2 and Fig 5, when the basic reproduction number is R_0 = 8.2 for the Omicron variant, we observe that a substantial ratio of the susceptible population (representing 84.48% of the total population) actively participates in the transmission process during the large-scale epidemic wave for k = 0.2812. In contrast to the homogeneous model, which assumes the entire population as being the initial number of susceptibles, the proposed model, incorporating the recruitment process of susceptible populations, significantly reduces the projected cumulative number of infections for the large-scale epidemic wave by 19.91% (i.e., 99.97% for the homogeneous model versus 80.06% for our model) for R_0 = 8.2.

Furthermore, if we take the point at which the effective reproduction number reaches unity as the time when the population attains herd immunity leading to the decline in new infections, we find that cumulative infections at that point amount to 49.61% of the population (i.e. the herd immunity) in our proposed model when $R_0 = 8.2$ and k = 0.2812. In comparison, the homogeneous model has already reached 86.97% of the population by that date when $R_0 = 8.2$. Similarly, it can be seen from Table 2 and Fig 5 that the peak sizes of daily new infections and the infectious class experience a substantial reduction, i.e. reduced by 29.61% and 47.94% for $R_0 = 8.2$, respectively, when spatio-temporal heterogeneity is incorporated.

In conclusion, with spatio-temporal heterogeneity considered, the peak size, final size and herd immunity corresponding to outbreaks of emerging infectious diseases are all significantly diminished as shown in Fig 6. The potential reason is that a significant proportion of susceptibles is not initially involved in the transmission process during the early phase, and when they



Fig 5. Solutions of the mechanistic model (model (4)) with three different values of k and the homogeneous model (model (1)) by varying the transmission rate β (corresponding to different variants of COVID-19). Here, we plotted the solution of S(t) of model (1) in the first panel. The corresponding effective reproduction numbers are shown in the last panel.

are gradually recruited into the transmission process, the former infected and recovered population, as a public immunity barrier, can offer cross-protection, thereby mitigating the risk of infection for the newly involved susceptibles. That is, the temporal heterogeneity of the susceptibles joining the transmission process contributes to the decrease of peak size, final size and herd immunity.

Additionally, upon examining the values presented in Table 2 across various variants of COVID-19, it becomes evident that the disparities in both the projected peak size and the final size between models with and without spatio-temporal heterogeneity become more pronounced as the basic reproduction number decreases from 8.2 to 2.2. This phenomenon can be more intuitively seen from Fig 6a. To illustrate this, when we fix k = 0.2812, the difference in attack rate projected by the two models with and without heterogeneity are 0.82, 0.71, 0.20 for $R_0 = 2.2$, 5, 8.2, respectively. In essence, this implies that for viruses (or other pathogens) with low transmissibility, the heterogeneity of the susceptible population can exert a significantly more discernible impact on the transmission dynamics of infectious diseases. This may also be attributed to the fact that as the reproduction number decreases, a considerably smaller proportion of the entire population become actual susceptibles during a single epidemic wave.

Prediction accuracy assessment

On 7 December 2022, China's National Health Commission announced major changes to the country's policies on COVID-19, marking a notable shift away from its zero-COVID policy. As a result, China experienced a significant COVID-19 epidemic wave [45,46] starting from that point. On 25 January 2023, the China CDC released a report detailing the national

	Indices	<i>k</i> = 0.15	k = 0.2812	k = 0.4	Model (1)		
		Unit: million					
$R_0 = 2.2$	Final size (Attack rate)	0.4295 (1.74%)	0.5393 (2.18%)	0.6928 (2.80%)	20.8775 (84.35%)		
	Peak of daily new infections (Rate to population)	0.0194 (0.08%)	0.0208 (0.08%)	0.0222 (0.09%)	1.0971 (4.43%)		
	Peak of infection class (Rate to the population)	0.0836 (0.34%)	0.0911 (0.37%)	0.0996 (0.40%)	4.6346 (18.73%)		
	Final reserve susceptible population $S_{r\infty}$	24.2011	24.0141	23.7476	3.8725		
	Herd immunity	1.02%	1.17%	1.34%	54.38%		
$R_0 = 5$	Final size (Attack rate)	0.9814 (3.97%)	6.9588 (28.12%)	15.6843 (63.37%)	24.5790 (99.31%)		
	Peak of daily new infections (Rate to the population)	0.0815 (0.33%)	0.0914 (0.37%)	0.3602 (1.46%)	4.4472 (17.97%)		
	Peak of infection class (Rate to the population)	0.2413 (0.98%)	0.4148 (1.68%)	1.7257 (6.97%)	11.8363 (47.82%)		
	Final reserve susceptible population $S_{r\infty}$	23.6269	16.3921	6.8531	0.1710		
	Herd immunity	1.82%	10.70%	35.67%	79.42%		
$R_0 = 8.2$	Final size (Attack rate)	6.0057 (24.27%)	19.8149 (80.06%)	22.9198 (92.61%)	24.7432 (99.97%)		
	Peak of daily new infections (Rate to the population)	0.1516 (0.61%)	0.7838 (3.17%)	1.5282 (6.17%)	8.1122 (32.78%)		
	Peak of infection class (Rate to the population)	0.3962 (1.60%)	3.5199 (14.22%)	6.1219 (24.73%)	15.3837 (62.16%)		
	Final reserve susceptible population $S_{r\infty}$	18.0593	3.8424	1.1676	0.0068		
	Herd immunity	7.30%	49.61%	63.67%	86.97%		

Table 2. Prediction of key epidemic indices for different scenarios using model (4).

Note: Attack rate = Final size/Total population. Herd immunity is defined here as the ratio of the total number of immune individuals (cumulative infections) to the population at the date when the effective reproduction number decreases to 1.

https://doi.org/10.1371/journal.pcbi.1012497.t002

situation on novel coronavirus infections. The report highlighted that the positive rate of infections detected by nucleic acid tests peaked on 22 December 2022, and by 23 January 2023, it had decreased to a very low level, signalling the end of the first epidemic wave after the easing of the zero-COVID policy. After the large scale outbreak, a recent study conducted four online surveys across 31 provincial-level administrative divisions in China to gather data on infection status, indicating that as of 7 February 2023, approximately 82.4% of the Chinese population had been infected [44]. Similarly, a large scale cohort study, by enrolling 14,744 participants in Shaanxi province of China, revealed a cumulative infection attack rate (IAR) of 84.7% for COVID-19 among the study participants after the easing of the zero-COVID policy [47]. Taking the survey data on the cumulative infection rate (attack rate) as the reference, the prediction accuracy of our model is defined as follows:

Prediction Accuracy =
$$100\% - \frac{|CI_{data} - CI_{model}|}{CI_{data}} * 100\%,$$

where CI_{data} is the real value from data and CI_{model} is the projected value from the model.

It follows from Table 2 that without considering the heterogeneity of the susceptibles, the attack rate projected by model (1) is 99.97% for the large-scale outbreak of the Omicron variant with a basic reproduction number of 8.2. Theoretical studies also indicate that the final epidemic size in the traditional *SIR* model exceeded 99% of the population when the basic

reproduction number reached 8.2 [48]. However, this outcome is markedly inconsistent with the actual situation reported in the aforementioned studies [44,47]. In contrast, our modelling framework, by involving the spatio-temporal heterogeneity, estimates the attack rate to be around 80.06% when $R_0 = 8.2$. Taking 82.4% as the reference, the prediction accuracy of our model reaches 97.2% while the prediction accuracy of the homogeneous model is 78.6%, i.e. resulting in an 18.6% improvement in the prediction accuracy.

Further, when the basic reproduction number is 8.2, the homogeneous model projects the peak size of daily new infections as 32.78%, which would be a huge number given the large scale outbreak of the Omicron variant in China and is out of line with the real case. Instead, our model projects a peak of around 3.17% of the population, aligning more closely with the estimated 4.7% of the population for the large-scale outbreak of the Omicron variant in Beijing city after the easing of the zero-COVID policy [45]. Moreover, an intuitive observation from Fig 5i and 5l reveals that the outbreak duration is approximately 70 days when we fix k = 0.2812 as the estimated value, whereas the homogeneous model projects a shorter duration of 30–40 days. Notably, from the announcement of major changes of control policy on 7 December 2022 to 23 January 2023 (the end of the epidemic wave), the first large-scale epidemic wave in China after the easing of the dynamic zero-COVID policies actually persisted for 77 days, which is also consistent with our projection.

In Fig 6, we further show the differences in the predictions of model (1) and model (4) by varying the basic reproduction number from 1.4 to 10. The horizontal bars represent the attack rate calculated from data, which results from the possible range of the basic reproduction number. Therefore, it can be intuitively seen that the prediction of the attack rate for model (4)



Fig 6. Comparison of four epidemic indices predicted by models (4) and (1), with the basic reproduction number varied between 1.4 and 10. The horizontal bars represent the reference values of the attack rate for the wild type with $R_0 \in (1.4, 3.9)$ [41], Delta variant with $R_0 \in (2.43, 5.11)$ [42], and Omicron variant strains with $R_0 \in (7.4, 9)$ [43], respectively. The reference value for Omicron is derived from survey data in [44]. The values for the wild type strain and the Delta variant were calculated by analyzing the rates of epidemics in Belarus under weak intervention measures. The data from 1 March to 1 October 2020, are used for the wild type strain, and data from 1 October 2020, to 20 December 2021, are used for the Delta variant.

https://doi.org/10.1371/journal.pcbi.1012497.g006

is much closer to the reference data marked as a purple bar in Fig 6a, indicating a significant improvement in prediction accuracy. Furthermore, we collected data on COVID-19 epidemics in the Republic of Belarus from Our World in Data (https://ourworldindata.org/coronavirus), where weak control interventions were implemented. The attack rate for the epidemics of the wild type strain and the Delta variant were around 0.8% and 6.6%, respectively, marked as rose red and green bars in Fig 6a. Using these data as a reference, we can intuitively see from Fig 6a the prediction accuracy for both the wild type strain and the Delta variant is also significantly improved in our model by incorporating heterogeneity.

In summary, by incorporating spatio-temporal heterogeneity into the classical homogeneous model, our model, in an efficient way, accurately depicts the epidemic curve of the first large-scale outbreak of the Omicron variant in China, including peak size, final size, herd immunity, and outbreak duration. Furthermore, the prediction accuracy is significantly enhanced compared to the homogeneous model.

Discussion

The presence of population heterogeneity is ubiquitous and represents a key factor in modelling infectious diseases, exerting a significant influence on transmission dynamics and epidemic outcomes [49–52]. Therefore, the development of effective modelling frameworks that account for spatial heterogeneity remains a central focus. Traditionally, patch models, dynamic network models, and diffusion models have been commonly employed to characterize the spatial spread of infectious diseases over time. However, their utility in conducting quantitative analyses of epidemic outbreaks, grounded in real-world data, is impeded by the inherent complexity and high dimensionality of these model structures in terms of the mechanistic modelling framework. This challenge has motivated us to devise a modelling framework characterized by a simplified structure, specifically designed to accommodate spatio-temporal heterogeneity. The objective is to render feasible the projection of key epidemic indices and enhance the overall performance of mathematical models under the influence of spatio-temporal heterogeneity, facilitating a more robust understanding of infectious disease dynamics.

As a solution, we introduced a pioneering modelling framework that incorporates spatiotemporal heterogeneity into traditional compartment models, thereby leveraging the inherent advantages of the compartmental modelling framework. This integration facilitates the precise quantification and prediction of critical epidemic parameters such as peak size, herd immunity, and final size of each epidemic wave. Our approach involved shaping the spread process of the susceptibles during a specific outbreak of the Omicron variant of COVID-19 in Shanghai, China, using two distinct methodologies. Employing the cutting-edge technology of deep learning, we trained our Universal Differential Equation (UDE) model and utilized the Markov Chain Monte Carlo (MCMC) method for calibrating the underlying mechanisms. The deep learning method played a crucial role in validating the assumptions embedded in the specific forms of the mechanistic model. Concurrently, the mechanistic model contributed to uncovering potential mechanisms in the recruitment process of susceptibles. Additionally, it addressed the limitations of the data-driven approach based on deep learning, particularly in terms of enhancing the generalization performance for conducting scenario analyses and predictions. This dual-method strategy synergistically combines the strengths of deep learning and mechanistic modelling to yield a comprehensive and robust framework for understanding and precisely predicting infectious disease dynamics.

A comparison of the estimated results for the transition rate shows that the deep learningbased data-driven approach is more flexible in capturing detailed information. Notably, there is a slight increase around day 65 in Fig 4c. The potential reason for this is that the reported numbers of infected cases suddenly rose during the decreasing phase of the epidemic wave (see Fig 4 for details), resulting in the increase of the recruitment rate of the susceptible population. This phenomenon also suggests the possibility that the diffusion of infections may lead to multiple instances of herd immunity, as the increased recruitment rate can cause the effective reproduction number to rise and exceed the threshold again. Even so, the mechanistic model can offer intuitive insights into the transmission and control of infectious diseases. The parameter k can be interpreted as the spatial diffusion speed in some contexts. In addition, we know that numerous factors, such as human behavioural changes, contact structures, and the evolution of the pathogen, influence the spread of infectious diseases. For simplicity, the impact of many of these factors on disease transmission can be represented by the transmission rate in epidemic models. For example, behavioural changes for self-protection can be reflected as a reduction in the transmission rate. Notably, the transmission rate parameter is incorporated into our mechanistic term for the recruitment rate of susceptibles. This offers an effective method for understanding and quantifying how various factors influence the recruitment process of susceptibles during an outbreak. Additionally, it helps to explain how control interventions affect the spatial spread of infections, providing an alternative approach to controlling infectious diseases. That is, a higher diffusion speed and larger transmission rate result in an increased recruitment rate of susceptibles, consequently leading to a higher attack rate, a peak in daily new infections, and a larger herd immunity, as can be intuitively seen in Fig 6. Therefore, interventions aimed at reducing the transmission capacity and spatial spread of the disease should be critical in controlling infectious diseases.

Through the incorporation of heterogeneity into the homogeneous model, we have observed a significant enhancement in prediction accuracy, with an approximately 18.6% improvement when assessing the cumulative attack rate during the large-scale outbreak of the Omicron variant in China, after the relaxation of the dynamic zero-COVID policy. Notably, when considering peak size, herd immunity, and the outbreak period, our proposed model exhibits projections that also closely align with real-world scenarios. Note that the outcomes of our study underscore that peak size, final size, and herd immunity in real-world scenarios are notably smaller than predictions derived from homogeneous models. Epidemic models are crucial tools that support decision-making in controlling infectious diseases. For instance, they help to determine the necessary preparation of medical resources during the peak of each epidemic wave or the vaccination coverage needed to achieve herd immunity. Therefore, accurate predictions of epidemic curves are essential for effective and cost-efficient outbreak management of infectious diseases. Consequently, the inclusion of heterogeneity becomes crucial for accurately estimating the requirements for medical resource preparation during large-scale outbreaks of emerging infectious diseases, as well as determining the necessary vaccination coverage. Our modelling framework presents a practical and effective method for precisely addressing these considerations. This highlights the importance of incorporating our modelling method into various compartmental modelling frameworks to enhance the prediction accuracy of epidemic models, ultimately improving decision-making in the control of infectious diseases.

Our novel framework of the mechanistic model (i.e. model (4)) adopts a remarkably simple method to incorporate heterogeneity into mathematical models. Consequently, this simplicity facilitates the extension of the modelling framework to consider various aspects of infectious disease transmission with spatio-temporal heterogeneity in the susceptible population. Moreover, this modelling structure opens the door to conducting more theoretical studies aimed at qualitatively analyzing the impact of heterogeneity on the transmission dynamics of infectious diseases and exploring optimal control strategies. In the light of these considerations, we contend that this study can serve as a foundation for the development and theoretical analysis of a class of models within the realm of modelling the heterogeneity of susceptible populations. The straightforward approach taken in our framework not only enhances its practical utility but also encourages further exploration and refinement, thereby contributing to the advancement of understanding and modelling infectious disease dynamics with spatio-temporal heterogeneity.

There are still several limitations to this study. Firstly, we only used one time-series of the epidemic data to train or calibrate the models, hence we appeal for the collection of more spatial data for calibrating our models, which would definitely help to increase the robustness of the results. Secondly, there is also another kind of heterogeneity (such as the age-structure of the population) which is ignored in the current study and how the different forms of heterogeneity reciprocally affect the transmission dynamics of infectious disease would be a very interesting issue. Finally, when we tried to generate the time series of the numbers of daily new infections, the frequency distribution of the time-interval from infection to report seems, potentially, to be bimodal, while our pre-set distributions were not bimodal. How the distribution of this time-interval will affect the results would also be an interesting issue. Furthermore, we did not consider demographic dynamics in our current model, as our focus is on a single outbreak of an emerging infectious disease. However, extending our modelling framework to incorporate the impact of demographic changes on the recruitment process of susceptibles, and subsequently on transmission dynamics, presents a very interesting topic for future research.

Author Contributions

Conceptualization: Biao Tang, Sanyi Tang, Yanni Xiao.

Data curation: Biao Tang, Xia Wang.

Formal analysis: Biao Tang, Kexin Ma, Yan Liu, Xia Wang.

Funding acquisition: Biao Tang, Sanyi Tang, Yanni Xiao.

Investigation: Biao Tang, Kexin Ma, Yan Liu, Xia Wang, Sanyi Tang, Yanni Xiao.

Methodology: Biao Tang, Kexin Ma, Yan Liu, Xia Wang, Sanyi Tang, Yanni Xiao.

Project administration: Biao Tang, Sanyi Tang, Yanni Xiao, Robert A. Cheke.

Resources: Biao Tang.

Software: Biao Tang, Kexin Ma, Yan Liu, Xia Wang.

Supervision: Biao Tang, Sanyi Tang, Yanni Xiao.

Validation: Biao Tang.

Visualization: Biao Tang, Kexin Ma, Yan Liu, Xia Wang.

Writing – original draft: Biao Tang, Kexin Ma.

Writing - review & editing: Biao Tang, Xia Wang, Sanyi Tang, Yanni Xiao, Robert A. Cheke.

References

- 1. Excler J-L, Saville M, Berkley S, Kim JH. Vaccine development for emerging infectious diseases. Nat Med. 2021; 27: 591–600. https://doi.org/10.1038/s41591-021-01301-0 PMID: 33846611
- Baker RE, Mahmud AS, Miller IF, Rajeev M, Rasambainarivo F, Rice BL, Infectious disease in an era of global change. Nat Rev Microbiol. 2022; 20: 193–205. https://doi.org/10.1038/s41579-021-00639-z PMID: 34646006

- Fauci AS, Morens DM. The perpetual challenge of infectious diseases. N Engl J Med. 2012; 366: 454– 461. https://doi.org/10.1056/NEJMra1108296 PMID: 22296079
- Stern AM. International efforts to control infectious diseases, 1851 to the present. JAMA. 2004; 292: 1474. https://doi.org/10.1001/jama.292.12.1474 PMID: 15383519
- Hellewell J, Abbott S, Gimma A, Bosse NI, Jarvis CI, Russell TW, et al. Feasibility of controlling COVID-19 outbreaks by isolation of cases and contacts. The Lancet Global Health. 2020; 8: e488–e496. https://doi.org/10.1016/S2214-109X(20)30074-7 PMID: 32119825
- Grundmann H, Hellriegel B. Mathematical modelling: a tool for hospital infection control. The Lancet Infectious Diseases. 2006; 6: 39–45. https://doi.org/10.1016/S1473-3099(05)70325-X PMID: 16377533
- Heesterbeek H, Anderson RM, Andreasen V, Bansal S, De Angelis D, Dye C, et al. Modeling infectious disease dynamics in the complex landscape of global health. Science. 2015; 347: aaa4339. <u>https://doi.org/10.1126/science.aaa4339</u> PMID: 25766240
- Abuelezam NN, Michel I, Marshall BD, Galea S. Accounting for historical injustices in mathematical models of infectious disease transmission: an analytic overview. Epidemics. 2023; 43: 100679. https:// doi.org/10.1016/j.epidem.2023.100679 PMID: 36924757
- McCallum H. How should pathogen transmission be modelled? Trends in Ecology & Evolution. 2001; 16: 295–300. https://doi.org/10.1016/s0169-5347(01)02144-9 PMID: 11369107
- Begon M, Bennett M, Bowers RG, French NP, Hazel SM, Turner J. A clarification of transmission terms in host-microparasite models: numbers, densities and areas. Epidemiol Infect. 2002; 129: 147–153. https://doi.org/10.1017/s0950268802007148 PMID: 12211582
- Lou Y, Salako RB. Mathematical analysis of the dynamics of some reaction-diffusion models for infectious diseases. Journal of Differential Equations. 2023; 370: 424–469. https://doi.org/10.1016/j.jde. 2023.06.018
- Hagenaars TJ, Donnelly CA, Ferguson NM. Spatial heterogeneity and the persistence of infectious diseases. Journal of Theoretical Biology. 2004; 229: 349–359. <u>https://doi.org/10.1016/j.jtbi.2004.04.002</u> PMID: 15234202
- Hu H, Nigmatulina K, Eckhoff P. The scaling of contact rates with population density for the infectious disease models. Mathematical Biosciences. 2013; 244: 125–134. https://doi.org/10.1016/j.mbs.2013. 04.013 PMID: 23665296
- Perkins TA, Scott TW, Le Menach A, Smith DL. Heterogeneity, Mixing, and the Spatial Scales of Mosquito-Borne Pathogen Transmission. Salathé M, editor. PLoS Comput Biol. 2013; 9: e1003327. https:// doi.org/10.1371/journal.pcbi.1003327 PMID: 24348223
- 15. Gao D, Ruan S. An SIS patch model with variable transmission coefficients. Mathematical Biosciences. 2011; 232: 110–115. https://doi.org/10.1016/j.mbs.2011.05.001 PMID: 21619886
- Wang Z, Xia C, Chen Z, Chen G. Epidemic propagation with positive and negative preventive information in multiplex networks. IEEE Trans Cybern. 2021; 51: 1454–1462. <u>https://doi.org/10.1109/TCYB.</u> 2019.2960605 PMID: 31940584
- Xiao D, Ruan S. Global analysis of an epidemic model with nonmonotone incidence rate. Mathematical Biosciences. 2007; 208: 419–429. https://doi.org/10.1016/j.mbs.2006.09.025 PMID: 17303186
- Hu Z, Ma W, Ruan S. Analysis of SIR epidemic models with nonlinear incidence rate and treatment. Mathematical Biosciences. 2012; 238: 12–20. https://doi.org/10.1016/j.mbs.2012.03.010 PMID: 22516532
- Vargas Soto JS, Kosiewska J, Muller LI, Grove D, Metts D, Wilber MQ. Correlated host movements can reshape spatio-temporal disease dynamics: modeling the contributions of space use to transmission risk using movement data. 2024. https://doi.org/10.1101/2024.04.16.589740
- Wilber MQ, Yang A, Boughton R, Manlove KR, Miller RS, Pepin KM, et al. A model for leveraging animal movement to understand spatio-temporal disease dynamics. Rohani P, editor. Ecology Letters. 2022; 25: 1290–1304. https://doi.org/10.1111/ele.13986 PMID: 35257466
- 21. Kermack WO, McKendrick AG. A contribution to the mathematical theory of epidemics. Proc R Soc Lond A. 1927; 115: 700–721. https://doi.org/10.1098/rspa.1927.0118
- Kermack WO, McKendrick AG. Contributions to the mathematical theory of epidemics. II.—The problem of endemicity. Proc R Soc Lond A. 1932; 138: 55–83. https://doi.org/10.1098/rspa.1932.0171
- Tang B, Xia F, Bragazzi NL, McCarthy Z, Wang X, He S, et al. Lessons drawn from China and South Korea for managing COVID-19 epidemic: Insights from a comparative modeling study. ISA Transactions. 2022; 124: 164–175. https://doi.org/10.1016/j.isatra.2021.12.004 PMID: 35164963
- Tang B, Zhou W, Wang X, Wu H, Xiao Y. Controlling multiple COVID-19 epidemic waves: an insight from a multi-scale model linking the behaviour change dynamics to the disease transmission dynamics. Bull Math Biol. 2022; 84: 106. https://doi.org/10.1007/s11538-022-01061-z PMID: 36008498

- 25. Tang B, Wang X, Li Q, Bragazzi NL, Tang S, Xiao Y, et al. Estimation of the transmission risk of the 2019-nCoV and its implication for public health interventions. JCM. 2020; 9: 462. <u>https://doi.org/10.3390/jcm9020462</u> PMID: 32046137
- Kucharski AJ, Russell TW, Diamond C, Liu Y, Edmunds J, Funk S, et al. Early dynamics of transmission and control of COVID-19: a mathematical modelling study. The Lancet Infectious Diseases. 2020; 20: 553–558. https://doi.org/10.1016/S1473-3099(20)30144-4 PMID: 32171059
- IHME COVID-19 Forecasting Team, Reiner RC, Barber RM, Collins JK, Zheng P, Adolph C, et al. Modeling COVID-19 scenarios for the United States. Nat Med. 2021; 27: 94–105. https://doi.org/10. 1038/s41591-020-1132-9 PMID: 33097835
- 28. Ge J, Kim KI, Lin Z, Zhu H. A SIS reaction–diffusion–advection model in a low-risk and high-risk domain. Journal of Differential Equations. 2015; 259: 5486–5509. https://doi.org/10.1016/j.jde.2015.06.035
- Choi W, Lin Z, Ahn I. SIS reaction–diffusion model with risk-induced dispersal under free boundary. Nonlinear Analysis: Real World Applications. 2022; 67: 103605. <u>https://doi.org/10.1016/j.nonrwa.2022</u>. 103605
- Rackauckas C, Ma Y, Martensen J, Warner C, Zubov K, Supekar R, et al. Universal differential equations for scientific machine learning. arXiv; 2021. http://arxiv.org/abs/2001.04385.
- Raissi M, Perdikaris P, Karniadakis GE. Physics-informed neural networks: A deep learning framework for solving forward and inverse problems involving nonlinear partial differential equations. Journal of Computational Physics. 2019; 378: 686–707. https://doi.org/10.1016/j.jcp.2018.10.045
- 32. Liu Z, Wang Y, Vaidya S, Ruehle F, Halverson J, Soljačić M, et al. KAN: Kolmogorov-Arnold Networks. arXiv; 2024.
- Song P, Xiao Y. Estimating time-varying reproduction number by deep learning techniques. J Appl Anal Comput. 2022; 12: 1077–1089. https://doi.org/10.11948/20220136
- 34. Pontryagin LS. Mathematical Theory of Optimal Processes. 1st ed. Routledge; 2018.
- Zhu W, Wen Z, Chen Y, Gong X, Zheng B, Liang X, et al. Age-specific transmission dynamics under suppression control measures during SARS-CoV-2 Omicron BA.2 epidemic. BMC Public Health. 2023; 23: 743. https://doi.org/10.1186/s12889-023-15596-w PMID: 37087436
- Shanghai Municipal Health Commission. Report of epidemic situation. [cited 30 Aug 2023]. <u>https://wsjkw.sh.gov.cn/yqtb/index.html</u>.
- Goldstein E, Dushoff J, Ma J, Plotkin JB, Earn DJD, Lipsitch M. Reconstructing influenza incidence by deconvolution of daily mortality time series. Proceedings of the National Academy of Sciences. 2009; 106: 21825–21829. https://doi.org/10.1073/pnas.0902958106 PMID: 20080801
- Shepp LA, Vardi Y. Maximum likelihood reconstruction for emission tomography. IEEE Trans Med Imaging. 1982; 1: 113–122. https://doi.org/10.1109/TMI.1982.4307558 PMID: 18238264
- Xia F, Xiao Y, Liu P, Cheke R A., Li X. Differences in how interventions coupled with effective reproduction numbers account for marked variations in COVID-19 epidemic outcomes. Mathematical Biosciences and Engineering. 2020; 17: 5085–5098. https://doi.org/10.3934/mbe.2020274 PMID: 33120541
- Haario H, Laine M, Mira A, Saksman E. DRAM: Efficient adaptive MCMC. Stat Comput. 2006; 16: 339– 354. https://doi.org/10.1007/s11222-006-9438-0
- Li Q, Guan X, Wu P, Wang X, Zhou L, Tong Y, et al. Early Transmission Dynamics in Wuhan, China, of Novel Coronavirus–Infected Pneumonia. N Engl J Med. 2020; 382: 1199–1207. https://doi.org/10.1056/ NEJMoa2001316 PMID: 31995857
- 42. Du Z, Hong H, Wang S, Ma L, Liu C, Bai Y, et al. Reproduction Number of the Omicron Variant Triples That of the Delta Variant. Viruses. 2022; 14: 821. https://doi.org/10.3390/v14040821 PMID: 35458551
- Liu Y, Rocklöv J. The effective reproductive number of the Omicron variant of SARS-CoV-2 is several times relative to Delta. Journal of Travel Medicine. 2022; 29: taac037. https://doi.org/10.1093/jtm/ taac037 PMID: 35262737
- 44. Fu D, He G, Li H, Tan H, Ji X, Lin Z, et al. Effectiveness of COVID-19 vaccination against SARS-CoV-2 Omicron variant infection and symptoms—China, December 2022–February 2023. China CDC Weekly. 2023; 5: 369–373. https://doi.org/10.46234/ccdcw2023.070 PMID: 37197449
- Leung K, Lau EHY, Wong CKH, Leung GM, Wu JT. Estimating the transmission dynamics of SARS-CoV-2 Omicron BF.7 in Beijing after adjustment of the zero-COVID policy in November–December 2022. Nat Med. 2023; 29: 579–582. https://doi.org/10.1038/s41591-023-02212-y PMID: 36638825
- Huang J, Zhao S, Chong KC, Zhou Y, Lu W, Fang F, et al. Infection rate in Guangzhou after easing the zero-COVID policy: seroprevalence results to ORF8 antigen. The Lancet Infectious Diseases. 2023; 23: 403–404. https://doi.org/10.1016/S1473-3099(23)00112-3 PMID: 36803917

- 47. Zhang T, Mi B, Shen M, Nian Y, Cao S, Liu J, et al. Exploring the transmission dynamics of the COVID-19 outbreaks after Dec. 2022 in Shaanxi province, China: analysis of baseline data from a large scale cohort. Epidemiology; 2023 Jan. https://doi.org/10.1101/2023.01.24.23284952
- Magal P, Seydi O, Webb G. Final size of an epidemic for a two-group SIR model. SIAM J Appl Math. 2016; 76: 2042–2059. https://doi.org/10.1137/16M1065392
- Hill AN, Glasser JW, Feng Z. Implications for infectious disease models of heterogeneous mixing on control thresholds. J Math Biol. 2023; 86: 53. <u>https://doi.org/10.1007/s00285-023-01886-9</u> PMID: 36884154
- Trauer JM, Dodd PJ, Gomes MGM, Gomez GB, Houben RMGJ, McBryde ES, et al. The importance of heterogeneity to the epidemiology of tuberculosis. Clinical Infectious Diseases. 2019; 69: 159–166. https://doi.org/10.1093/cid/ciy938 PMID: 30383204
- Buckee C, Noor A, Sattenspiel L. Thinking clearly about social aspects of infectious disease transmission. Nature. 2021; 595: 205–213. https://doi.org/10.1038/s41586-021-03694-x PMID: 34194045
- Mishra S, Kwong JC, Chan AK, Baral SD. Understanding heterogeneity to inform the public health response to COVID-19 in Canada. CMAJ. 2020; 192: E684–E685. https://doi.org/10.1503/cmaj.201112 PMID: 32493741